Author's response to reviews

Title: Identifying factors which enhance capacity to engage in clinical education among podiatry practitioners: an action research project

Authors:

Sally Abey (sabey@plymouth.ac.uk)
Susan Lea (susan.lea@kcl.ac.uk)
Susan Callaghan (L.Callaghan@plymouth.ac.uk)
Steve Shaw (S.Shaw@plymouth.ac.uk)
Debby Cotton (D.Cotton@plymouth.ac.uk)

Version: 8
Date: 2 November 2015

Author's response to reviews: see over
2nd November 2015

FACULTY OF HEALTH AND HUMAN SCIENCES
Plymouth University
Marjon Campus
PAHC
Derriford Road
Plymouth
PL6 9BH

Dear Editor

Re: Identifying factors which enhance capacity to engage in clinical education among podiatry practitioners: an action research project

Thank you for accepting the above paper and for your kind assistance throughout the submission process.

Yours faithfully

Sally Abey PhD